Zoonotic *Enterocytozoon bieneusi* genotypes in free-ranging and farmed wild
 ungulates in Spain

4	Mr. Alejandro Dashti ^{1,†} , Dr. Mónica Santín ^{2,†} , Ms. Pamela C. Köster ¹ , Ms. Begoña Bailo ¹ ,
5	Ms. Sheila Ortega ¹ , Ms. Elena Imaña ¹ , Dr. Miguel Ángel Habela ³ , Dr. Antonio Rivero-
6	Juarez ⁴ , Dr. Joaquin Vicente ⁵ , WE&H group ^{6,‡} , Dr. Maria C. Arnal ⁷ , Dr. Daniel
7	Fernández de Luco ⁷ , Dr. Patrocinio Morrondo ⁸ , Dr. José A. Armenteros ⁹ , Dr. Ana
8	Balseiro ^{10,11} , Dr. Guillermo A. Cardona ¹² , Dr. Carlos Martínez-Carrasco ¹³ , Dr. José
9	Antonio Ortiz ¹⁴ , Dr. Rafael Calero-Bernal ¹⁵ , Dr. David Carmena ^{1*} , Dr. David González-
10	Barrio ^{1*}
11	[†] These first authors contributed equally to this article.
12	[‡] Members of the group are listed at the end of the article.
13	
14	¹ Parasitology Reference and Research Laboratory, Spanish National Centre for Microbiology
15	Health Institute Carlos III, Majadahonda, Madrid, Spain
16	² Environmental Microbial and Food Safety Laboratory, Agricultural Research Service, United
17	States Department of Agriculture, Beltsville, Maryland, USA
18	³ Department of Animal Health, Veterinary Sciences Faculty, Extremadura University, Caceres,
19	Spain
20	⁴ Infectious Diseases Unit, Maimonides Institute for Biomedical Research (IMIBIC), University
21	Hospital Reina Sofía, University of Córdoba, Córdoba, Spain
22	⁵ SaBio Group, Institute for Game and Wildlife Research, IREC (UCLM-CSIC-JCCM), Ciudad
23	Real, Spain
24	⁶ Wildlife Ecology & Health Group (WE&H), Wildlife Environmental Pathology Service
25	(SEFaS), Department of Animal Medicine and Surgery, Autonomous University of Barcelona,
26	Bellaterra, Spain

- ⁷ Department of Animal Pathology, Veterinary Faculty, University of Zaragoza, Zaragoza, Spain
- ⁸ INVESAGA Group, Department of Animal Pathology, Faculty of Veterinary, University of
- 29 Santiago de Compostela, Lugo, Spain
- ⁹ Council of Development, Territory Planning and the Environment of the Principado de Asturias,
- 31 Oviedo, Spain
- 32 ¹⁰ Animal Health Department, Veterinary School, University of León, León, Spain
- ¹¹ Animal Health Department, Mountain Livestock Institute (CSIC-University of León), León,
 Spain
- 35 ¹² Livestock Laboratory, Regional Government of Álava, Vitoria-Gasteiz, Spain
- 36 ¹³ Animal Health Department, University of Murcia, Regional Campus of International
- 37 Excellence "Campus Mare Nostrum", Espinardo, Murcia, Spain
- ¹⁴ Medianilla S.L., Department of Veterinary and Research. Benalup-Casas Viejas, Spain
- ¹⁵ SALUVET, Department of Animal Health, Faculty of Veterinary, Complutense University of
- 40 Madrid, Madrid, Spain
- 41
- 42 Corresponding authors: Parasitology Reference and Research Laboratory, National
- 43 Centre for Microbiology, Ctra. Majadahonda-Pozuelo Km 2, 28220 Majadahonda,
- 44 Madrid, Spain.
- 45 <u>dacarmena@isciii.es; dgonzalezbarrio@gmail.com</u>
- 46
- 47 Keywords
- 48 Enterocytozoon bieneusi; wild ungulates; molecular diversity; Spain

- 50 **Conflict of Interest**
- 51 The authors have declared no conflict of interest.
- 52
- 53

54 Abstract

Microsporidia comprises a diverse group of obligate, intracellular, and spore-forming 55 56 parasites that infect a wide range of animals. Among them, Enterocytozoon bieneusi is the most frequently reported species in humans and other mammals and birds. Data on 57 the epidemiology of E. bieneusi in wildlife is limited. Hence, E. bieneusi was investigated 58 in eight wild ungulate species present in Spain (genera Ammotragus, Capra, Capreolus, 59 Cervus, Dama, Ovis, Rupicapra, and Sus) by molecular methods. Faecal samples were 60 collected from free-ranging (n = 1058) and farmed (n = 324) wild ungulates from five 61 Spanish bioregions. The parasite was detected only in red deer (10.4%, 68/653) and wild 62 boar (0.8%, 3/359). Enterocytozoon bieneusi infections were more common in farmed 63 (19.4%, 63/324) than in wild (1.5%, 5/329) red deer. Eleven genotypes were identified in 64 red deer, eight known (BEB6, BEB17, EbCar2, HLJD-V, MWC d1, S5, Type IV, and 65 Wildboar3) and three novel (DeerSpEb1, DeerSpEb2, and DeerSpEb3) genotypes. Mixed 66 genotype infections were detected in 15.9% of farmed red deer. Two genotypes were 67 identified in wild boar, a known (Wildboar3) and a novel (WildboarSpEb1) genotypes. 68 69 All genotypes identified belonged to *E. bieneusi* zoonotic Groups 1 and 2. This study provides the most comprehensive epidemiological study of E. bieneusi in Spanish 70 71 ungulates to date, representing the first evidence of the parasite in wild red deer populations worldwide. Spanish wild boars and red deer are reservoir of zoonotic 72 genotypes of E. bieneusi and might play an underestimated role in the transmission of this 73 74 microsporidian species to humans and other animals.

75

76 Lay Summary

The fungal-related intracellular parasite *Enterocytozoon bieneusi* is a worldwide public
health and veterinary problem. Here we demonstrated that it was present in wild boar,

and wild and farmed red deer in Spain, with genotypes potentially capable of infectinghumans, posing a public health risk.

81

82 Introduction

Microsporidia is a diverse group of obligate, intracellular, and spore-forming parasites 83 related to fungi that infect a wide range of vertebrate and invertebrate hosts.¹ At least 220 84 genera and 1,700 species of Microsporidia have been described so far, of which 17 species 85 are able to infect humans. Among them, Enterocytozoon bieneusi is regarded as the most 86 frequent species causing human microsporidiosis.² Enterocytozoon bieneusi is primarily 87 identified in immunocompromised (including HIV+) patients associated with chronic 88 diarrhoea and, to a lesser extent, extra-intestinal clinical manifestations.^{3,4} However, its 89 presence has been increasingly reported in apparently healthy individuals in recent 90 years.^{5–8} The routes of transmission of this parasite have not been fully elucidated yet; it 91 92 is likely that the major route is via faecal-oral transmission of spores through direct 93 contact with infected animals (including humans), or by ingestion of contaminated food 94 and water. Indeed, E. bieneusi has been identified as the causative agent of a foodborne outbreak of microsporidiosis in Denmark.9 95

To date, over 600 *E. bieneusi* genotypes have been identified based on the analysis of the ITS region of the parasite.¹⁰ These genetic variants have been allocated into 11 phylogenetic major groups, which Group 1 and 2 containing most genotypes with zoonotic potential, whereas the remaining (groups 3–11) include mostly host-adapted genotypes associated to specific animals.¹¹

101 Along human history, wildlife has been an important source of infectious diseases 102 for humans.¹² Currently, zoonotic pathogens with a wildlife reservoir constitute a major 103 public health problem, affecting all continents. Several viruses, bacteria and parasites have been able to cross the host species (e.g. fox, red deer, wild boar) barrier to emerge
as zoonoses.¹³

106 During the last few decades, wild ungulates species in Europe have spatially expanded due to different factors: the intensification of game management practices, 107 human depopulation of rural areas, changes in land use, introduction of individuals 108 outside their native range, or reintroductions of endangered species.^{14–16} Data on the 109 110 epidemiology of E. bieneusi in ungulates species are limited. Most of the studies conducted globally have focused on the presence of this microsporidia in farmed or 111 captive ungulate species,¹¹ but occurrence and molecular data in wild ungulates remain 112 113 largely unknown. Enterocytozoon bieneusi has been documented in wild ungulate species of the genera Axis, Capreolus, Cervus, Dama, Hydropotes, Kobus, Muntiacus, 114 115 Odocoileus, Rangifer, Rusa, and Sus at prevalence rates ranging from 0-42% in farmed 116 animals, and from 0-54% in free-living animals, with sporadic cases identified in captive animals at zoological institutions (Table 1). Most of the E. bieneusi genotypes identified 117 in those hosts belong to the zoonotic Groups 1 and 2, but others are included in host-118 adapted Group 3 and Group 8^{11} (Table 1). Only three studies have reported *E. bieneusi* in 119 free-living wild boar populations, with infection rates ranging from 2–14% (Table 1). All 120 121 available data on this host come from studies conducted in European (Austria, Czech Republic, Poland, Slovakia, and Spain) and Asian (South Korea) countries. These studies 122 revealed a limited E. bieneusi genetic diversity in wild boars, being EbpA and EbpC 123 124 (Group 1) the most prevalent genotypes described in this host (Table 1). There are no reports of E. bieneusi in wild red deer; all studies in this host were conducted in farmed 125 or captive animals in China (Table 1). Infection rates ranged from 8-38% with five E. 126 bieneusi genotypes identified (BEB6, JLD-IV, JLD-XIII, HLJD-V, and HLJD-VI). 127

128

INSERT HERE TABLE 1

130 In Spain, wild ungulate species including wild boar and red deer are well-known suitable hosts of zoonotic infectious pathogens such as Mycobacterium bovis, hepatitis E 131 virus, and *Coxiella burnetii*.^{39–42} However, little information is currently available about 132 the epidemiology of E. bieneusi in wild ungulates in the country. Just a single study has 133 previously reported E. bieneusi in Spanish wild boars, but this survey was conducted only 134 at regional scale.³⁸ Hence, this study was carried out to determine the prevalence, genetic 135 diversity, and zoonotic potential of E. bieneusi in a large population of free-ranging and 136 farmed wild ungulates from different Spanish regions to gain national representativeness. 137

(References 17–38)

138

Materials and methods

140 Study area and sampling strategy

Between 1999 and 2021, a retrospective nationwide survey was performed. Faecal samples from the eight wild ungulate species present in Spain: Barbary sheep (*Ammotragus lervia*), Iberian wild goat (*Capra pyrenaica*), roe deer (*Capreolus capreolus*), red deer (*Cervus elaphus*), fallow deer (*Dama dama*), mouflon (*Ovis aries musimon*), Southern chamois (*Rupicapra pyrenaica*), and wild boar (*Sus scrofa*), were collected throughout the five bioregions (BRs, see below) of mainland Spain (Table 1, Fig. 1).

Based on landscape structure, major ecosystems, game management practices, and socio-political aspects, the Spanish Wildlife Disease Surveillance Scheme splits mainland Spain into five different BRs (Fig. 1) sharing similar epidemiological features.⁴³ BR1 comprises the Northern areas of temperate Atlantic climate with almost no game management; meanwhile, the remaining BRs present a Mediterranean climate with an increasing drought gradient from BR2 to BR4. In the Mediterranean BRs, game

management is not the norm except for BR3 and the Southwest of BR5, where the highly 154 155 productive savannah-like or oak forest landscapes are frequently profited for large game 156 production. Mountain habitats are more dominant in BRs 1, 2, and 5, while cereal plains are predominant in BR4. This zoning has been previously exploited to facilitate disease 157 surveillance efforts in wild ungulates in Spain.^{39,44–47} From each sampling site, that is, 158 hunting states or game reserves (n = 65; Fig. 1) selected by simple random sampling 159 160 throughout the study area, the animals (15-20 whenever possible) were also randomly 161 sampled.

All animals were legally harvested by hunters or culled as part of population 162 163 control programmes on game reserves. Faecal samples were collected directly from the rectum of each animal during field necropsies after hunting using disposable gloves and 164 placed in individual sterile tubes with records of the date, location, and host. Collected 165 166 samples were transported in cooled boxes to each participating institution responsible for the sampling and stored at -20 °C. Aliquots of these faecal samples were shipped to the 167 Spanish National Centre for Microbiology, Majadahonda (Spain) for subsequent 168 molecular analyses. 169

170 Aliquots of faecal samples from farmed red deer belonging to a semi-extensively bred red deer population located in southern Spain were obtained from a previous work.⁴⁸ 171 The deer were semi-extensively bred in a forest-shrub prairie habitat divided into different 172 plots by high-wire fencing. The animals were kept in separate batches according to their 173 174 sex and productive status. They were kept within large fenced (6-8 ha) enclosures in batches of 60-80 reproductive females; the males were kept in separate enclosures. The 175 animals were identified with individual ear tags. Faecal material was collected directly 176 from the rectum using sterile disposable latex gloves during health veterinary inspections. 177

179 DNA extraction and purification

180 Genomic DNA was isolated from about 200 mg of each faecal specimen of wild ungulate 181 origin by using the QIAamp DNA Stool Mini Kit (Qiagen, Hilden, Germany) according 182 to the manufacturer's instructions, except that samples mixed with InhibitEX buffer were 183 incubated for 10 min at 95 °C. Extracted and purified DNA samples were eluted in 200 184 μ l of PCR-grade water and kept at 4 °C until further molecular analysis.

185

186 PCR and sequence analysis

To identify E. bieneusi, a nested PCR protocol was used to amplify the ITS region as well 187 as portions of the flanking large and small subunit of the ribosomal RNA gene as 188 previously described.⁴⁹ The outer (EBITS3 and EBTIS4) and inner (EBITS1 and 189 190 EBITS2.4) primer sets were used to generate PCR products of 435 and 390 bp, respectively. Negative and positive controls were included in every PCR run. The 191 amplicons of the second PCR were examined on 2% D5 agarose gels stained with 192 Pronasafe (Conda, Madrid, Spain). All amplicons of the expected size were directly 193 sequenced in both directions with the internal primer pair in 10 μ l reactions using Big 194 DyeTM chemistries and an ABI 3730xl sequencer analyser (Applied Biosystems, Foster 195 196 City, CA). Raw sequences were examined with Chromas Lite version 2.1 software (http://chromaslite.software.informer.com/2.1) to generate consensus sequences. These 197 sequences were compared with reference sequences deposited at the National Center for 198 Biotechnology Information the BLAST 199 (NCBI) using tool 200 (http://blast.ncbi.nlm.nih.gov/Blast.cgi). The established nomenclature system based in ITS nucleotide sequence was used to determine E. bieneusi genotypes.⁵⁰ Sequences 201 202 generated in the present study were deposited in the GenBank public repository database under accession numbers ON819430-ON819442. 203

When E. bieneusi mixed genotype infection within a specimen was suspected from the 206 chromatogram sequence traces, the secondary PCR products were cloned using the TOPO 207 TA cloning kit (Invitrogen Corp., Carlsbad, CA, USA). Transformants (eight clones from 208 each specimen) were selected, PCR-amplified, and sequenced in both directions using 209 210 M13 forward and reverse primers. Briefly, amplicons were purified using Exonuclease I/Shrimp Alkaline Phosphatase (ExoSAP-IT Express, Affymetrix Inc., Santa Clara, CA, 211 212 USA), and sequenced in both directions using primers utilized for PCR screening in 10 μ l reactions, Big DyeTM chemistries, and an ABI 3130 sequencer analyser (Applied 213 214 Biosystems).

215

216 Sequence and phylogenetic analysis

Sequence chromatograms of each strand were aligned and examined with Lasergene software (DNASTAR, Inc., Madison, WI, USA). Sequences obtained in this study as well as *E. bieneusi* sequences previously identified in livestock, wildlife, and companion animals in Spain and appropriate reference sequences retrieved from GenBank were aligned with the Clustal W algorithm. Phylogenetic analysis was performed using the Neighbour-Joining (NJ) method, and genetic distance was calculated with the Kimura parameter-2 model using MEGA X.^{51,52}

224

225 Statistical analysis

The Pearson's χ^2 test was used to assess differences in *E. bieneusi* occurrence rates according to host species, population type (wild *versus* farmed), and bioregion (BR1-5)

of origin. Analyses were carried out using the R Statistical Package version 2.15.3.⁵³

229

230 **Results**

231 Occurrence of *E. bieneusi*

A total of 1382 faecal samples were collected from wild ungulates (76.6%, 1058/1382) 232 and farmed red deer (23.4%, 324/1382) from different Spanish regions during the period 233 234 1999–2021 (Supplementary Table 1). Overall, 5.1% (71/1382; 95% CI: 4.0–6.4) of the faecal samples from ungulates analysed were positive for E. bieneusi by PCR. Parasite 235 236 DNA was detected in red deer (10.4%, 68/653; 95% CI: 8.3–12.9) and wild boars (0.8%, 3/359; 95% CI: 0.3–2.4), but not in fallow deer (0/96; 95% CI: 0.00–3.8), roe deer (0/93; 237 238 95% CI: 0.00-3.9), mouflons (0/10; 95% CI: 0.00-27.7), Iberian wild goat (0/89; 95% CI: 0.00-4.1), Barbary sheep (0/20; 95% CI: 0.00-16.1), or Southern chamois (0/62; 239 240 95% CI: 0.00-5.8) (Table 2, Fig. 1). Among the red deer populations, the occurrence of E. bieneusi was statistically higher in farmed red deer (19.4%, 63/324) than in wild red 241 deer (1.5%, 5/329) [χ^2 (1, n = 653) = 56.2, P < 0.001]. 242

Regarding spatial distribution by bioregion, *E. bieneusi* was detected in both red deer and wild boar in the three BR regions, BR5 (8.7%, 66/748), BR3 (1.2%, 4/335), and BR2 (1.2%, 2/164), without statistically significant differences (Table 2, Fig. 1).

The full dataset of this study showing sampling, epidemiological, diagnostic, andmolecular data can be found in Supplementary Table 2.

248

249 Molecular characterization of *E. bieneusi*

In wild boars, sequence analysis of the ITS revealed the presence of two genotypes, a previously reported genotype in wild boar (Wildboar3) and a novel genotype (named WildboarSpEb1) (Table 3). WildboarSpEb1 differed by a single nucleotide polymorphism (SNP) at ITS region with genotype EbpA (AF076040) at nucleotide site 254 113 (C→T). Wildboar3 was detected in one animal and WildboarSpEb1 in two animals, 255 one each in BR3 and BR5 (Supplementary Table 1).

256 In red deer, analysis of the nucleotide sequences at the ITS region revealed a high genetic diversity with 11 distinct E. bieneusi genotypes circulating alone or in 257 combination. Out of the 11 genotypes, eight were known genotypes (HLJD-V, BEB6, 258 BEB17, MWC d1, S5, EbCar2, Type IV, and Wildboar3) and three novel genotypes 259 260 (named DeerSpEb1, DeerSpEb2, and DeerSpEb3) (Table 3). Mixed infections involving two or more genotypes were identified in 15.9% (10/63) of the farmed red deer samples 261 analysed (Supplementary Table 1). DeerSpEb1 differed by a single SNP from genotype 262 263 FJL (MK357781) at nucleotide site 78 (G \rightarrow T); DeerSpEb2 differed by a SNP from genotype LND-I (MN056217) at nucleotide site 144 ($A \rightarrow G$); and DeerSpEb3 differed by 264 265 a SNP from nucleotide sequence with no genotype information isolated from a Père 266 David's deer (MG703260) at nucleotide site 144 ($A \rightarrow G$).

HLJD-V was the most prevalent genotype identified in red deer (52.9%, 36/68),
followed by DeerSpEb2 (10.3%, 7/68), Wildboar3 (5.9%, 4/68), and DeerSpEb1 (5.9%,
4/68) (Table 3). Genotypes EbCar2, BEB17, S5, and Type IV were only observed in wild
red deer, whereas genotypes HLJD-V, BEB6, MWC_d1, Wildboar3, DeerSpEb1,
DeerSpEb2, and DeerSpEb3 were found in farmed red deer only (Table 3). Regarding the
bioregion of origin, EbCar2 was only found in wild red deer populations from BR2,
BEB17 and S5 in BR33 and Type IV in BR5 (Supplementary Table 3).

274

275 Phylogenetic analysis

Phylogenetic analysis of ITS sequences using the NJ method demonstrated that novel
genotypes belonged to groups of *E. bieneusi* that contain zoonotic genotypes. DeerSpEb1

and WildboarSpEB1 clustered within the Group 1, and DeerSpEb2 and DeerSpEb3
within the Group 2 (Fig. 2).

280

281 Discussion

In Spain, information on the occurrence and molecular diversity of E. bieneusi in wildlife 282 is limited. Recent studies have identified this microsporidian species in wild boars and 283 Iberian pigs sharing the same habitat,³⁸ wild and domestic carnivores,^{54–56}, lagomorphs,⁵⁷ 284 urban pigeons,^{58,59} and wild micromammals.⁶⁰ In our study, *E. bieneusi* was identified in 285 10.4% and 0.8% of the investigated red deer and wild boar populations, respectively. Of 286 note, infection rates were significantly higher in farmed (19.4%) than in wild (1.5%) red 287 288 deer. This large discrepancy can be explained by two reasons: i) farmed animals confined 289 in limited enclosures have higher group sizes, densities, and interaction rates (all favouring parasite transmission) than free-living animals, and ii) the surveyed deer farm 290 291 features a great faunal biodiversity and is located in the European-African migration 292 route, factors that promote inter-species parasite transmission. Indeed, higher nematodal parasite burdens have been previously found in farmed deer raised at high densities than 293 in wild deer populations in Argentina.⁶¹ Similarly, a direct relationship between host 294 density and parasite burdens has been demonstrated in white-tailed deer in the USA⁶² and 295 in wild cervids (Cervus elaphus and Dama dama) in Spain.63 296

To date, *E. bieneusi* has been reported in farmed and captive (zoo) red deer in China^{26,27} (Table 1). Present survey reports for the first time the presence of *E. bieneusi* in wild red deer populations worldwide. The infection rate found in farmed red deer (19.4%) was similar to that reported in farmed red deer in China (20.0%),²⁷ but lower than that identified in farmed and zoo animals (37.5%) in the same country.²⁶ Although *E. bieneusi* has also been reported in wild roe deer in Korea,²² we did not detect this

parasite in roe deer in this study. This could be related to the relatively low number of roe 303 304 deer samples collected in the study (n = 93) coupled with fact that a low infection rate was observed in wild red deer in this study (1.5%). This will also explain the negative 305 results for presence of E. bieneusi for the other ungulates including in the study: fallow 306 deer (n = 96), mouflons (n = 10), Iberian wild goat (n = 90), Barbary sheep (n = 20), and 307 Southern chamois (n = 62). Clearly, more studies including higher number of samples 308 309 appear to be required to investigate this parasite in wild populations due to the expected low prevalence. 310

Enterocytozoon bieneusi was detected at low infection rates (0.8%, 3/359) in the investigated wild boar population. This figure was slightly lower than that (2.1%) recently reported in wild boars in Southern Spain.³⁸ Comparatively higher occurrence rates (8– 14%) have been documented in Central European countries including Austria, Czech Republic, Poland, and Slovak Republic³⁴ and in South Korea (3%).³⁷

An interesting contribution of this study is the demonstration that red deer are 316 suitable hosts for a very large diversity (n = 11) of *E. bieneusi* genotypes. Besides the 317 318 eight already known genotypes (BEB6, BEB17, EbCar2, HLJD-V, MWC d1, S5, and Wildboar3), three novel genotypes named DeerSpEb1, DeerSpEb2, and DeerSpEb3 were 319 320 additionally described. In previous studies, only genotypes BEB6, HLJD-V, HLJD-VI, JLD-IV, and JLD-XIII were found circulating in red deer populations (see Table 1). 321 Therefore, this study constitutes the first report of genotypes BEB17, EbCar2, S5, 322 Wildboar3, DeerSpEb1, DeerSpEb2, and DeerSpEb3 in this host. Furthermore, we 323 identified mixed infections in farmed red deer involving two or more genotypes in a single 324 faecal sample, suggesting that these infections were common in semi-captive animals due 325 to increased contact among animals or to cross-species transmission from synathropic 326 hosts infected by the pathogen. This finding may also have unforeseen public health 327

consequences, as farm workers may be more exposed to *E. bieneusi* infections during the
handling of these animals or their manure. In this farm, management interventions
(sanitary issues, weaning, reposition, and artificial insemination) were limited to two to
four times per year to minimise the risk of animal stress.³⁹

Genotype HLJD-V was the most prevalent E. bieneusi genotype identified in red 332 deer. To date, this genetic variant had only been detected in cervids including red deer, 333 fallow deer, sika deer, Chinese water deer, and Père David's deer in China.^{18,24,27}. In 334 addition, genotype Wildboar3 has been previously reported in wild foxes and badgers in 335 Spain,⁵⁵ farmed foxes and raccoons in China,^{64–66}, wild boars in Central Europe,³⁴ and 336 wild raccoons in Poland.⁶⁷. Genotype BEB17 has been only reported in cattle in Brazil.⁶⁸ 337 Therefore, this is the second report of this E. bieneusi genotype worldwide. In Spain, 338 genotype BEB6 was previously detected in domestic dogs from the northern area of the 339 country.⁵⁶ This genotype is commonly seen in cervids (see Table 1) and other animals 340 including human and non-human primates, alpacas, horses, cattle, cats, sheep, goats, and 341 342 birds, suggesting that BEB6 has loose host specificity and, therefore, has zoonotic 343 potential.¹¹ Genotype MWC d1 was first reported in wild Sambar deer in Australia²⁰ and subsequently described in wild Père David's deer in China.²⁴ In the present study, a single 344 red deer was found infected with genotype S5. This E. bieneusi genetic variant has been 345 reported in wild badgers in Spain⁵⁵ and in four HIV-positive adults in Malawi,⁶⁹ 346 suggesting that this genotype has zoonotic potential and cross-transmission between 347 humans and animals is possible. Additionally, this is the second report of genotype 348 EbCar2, a variant previously found infecting badgers in Spain and Poland.^{55,70} Finally, 349 Type IV was observed in a single sample from free-ranging wild red deer. Although this 350 is the first description of this genotype in this host, Type IV has been commonly reported 351

in humans and numerous hosts including non-human primates, bovids, other cervid
 species, rodents, cats, birds, and domestic dogs¹¹.

Two *E. bieneusi* genotypes were identified in wild boars including known Wildboar3 genotype and novel WildboarSpEb1 genotype. Only genotypes EbpA and PigSpEb1 had been previously described in Spanish wild boars,³⁸ so this is the first description of Wildboar3 in this host in Spain. Of note Wildboar3 genotype was first described in wild boars from Czech Republic and Poland³⁴ and subsequently identified in other European wildlife species including introduced raccoon dogs in Poland and Germany,⁶⁷ and badgers and red foxes in Spain.⁵⁵

361 In conclusion, this large molecular-based epidemiological survey provides firsttime nationwide data on the presence and genetic diversity of E. bieneusi in wild ungulate 362 populations in Spain. Major contributions of the survey include i) first report of E. 363 364 bieneusi in wild red deer worldwide, ii) first description of this pathogen in farmed red deer in Spain, iii) confirmation that all known and novel E. bieneusi genotypes described 365 belonged to the zoonotic Groups 1 and 2, and iv) expansion of the known host range for 366 certain E. bieneusi genotypes. The relatively common finding of zoonotic E. bieneusi 367 genetic variants in wild boars and red deer - the two more abundant and widely distributed 368 369 wild ungulate species - pose a public health risk for individuals (e.g. veterinarians, farmers, hunters) in close contact with these animals or their manure that should not be 370 underestimated. Overall, these results expand our current knowledge on the epidemiology 371 and public veterinary health relevance of E. bieneusi. 372

373

374 Supplementary Material

375 Supplementary data are available at MMYCOL online.

377 Acknowledgment

D.G.-B. is the recipient of a Sara Borrell Research Contract (CD19CIII/00011) funded by
the Spanish Ministry of Science, Innovation and Universities. A.D. is the recipient of a
PFIS contract (FI20CIII/00002) funded by the Spanish Ministry of Science and
Innovation and Universities.

382

383 Funding

This work was supported by Health Institute Carlos III (ISCIII), Spanish Ministry of Economy and Competitiveness, under project PI19CIII/00029. Sample collection in Catalonia was supported by the Council of Barcelona through contracts 13/051, 15/0174, 16/0243 and 16/0243-00-PR/01, from the Spanish Ministry of Economy and Competitiveness under research grants CGL2012- 40043-C02-01, CGL2012-40043-C02-02, and from the Spanish Ministry of Science and Innovation under research grant CGL2016-80543-P PID2020-115046GB-I00, and

391

392 Wildlife Ecology & Health Group (WE&H)

The following investigators (in alphabetic order) participated in the Wildlife Ecology &
Health Group (WE&H): Carles Conejero, Carlos González-Crespo, Cristina Garrido,
Diana Gassó, Diana Andrea Murillo, Emmanuel Serrano, Gregorio Mentaberre, Irene
Torres-Blas, Josep Estruch, Josep Pastor, Jorge Ramón López-Olvera, María EscobarGonzález, Marta Valldeperes, Montse Mesalles, Omar López, Raquel Álvarez, Rafaela
Cuenca, Roser Velarde, Santiago Lavín.

399

400 **References**

- Stentiford GD, Becnel JJ, Weiss LM et al. Microsporidia-emergent pathogens in the
 global food chain. *Trends Parasitol.* 2016; 32: 657.
- 403 2. Han B, Pan G, Weiss LM. Microsporidiosis in humans. *Clin Microbiol Rev.* 2021; 34:
 404 e0001020.
- 405 3. Halánová M, Valenčáková A, Jarčuška P et al. Screening of opportunistic
 406 *Encephalitozoon intestinalis* and *Enterocytozoon bieneusi* in immunocompromised
 407 patients in Slovakia. *Cent Eur J Public Health*. 2019; 27: 330–334.
- 408 4. Hartskeerl RA, Schuitema AR, van Gool T et al. Genetic evidence for the occurrence
 409 of extra-intestinal *Enterocytozoon bieneusi* infections. *Nucleic Acids Res.* 1993; 21:
- 410 4150.
- 5. Sak B, Brady D, Pelikánová M et al. Unapparent microsporidial infection among
 immunocompetent humans in the Czech Republic. J Clin Microbiol. 2011; 49: 1064–
 1070.
- 414 6. Sak B, Kváč M, Kučerová Z et al. Latent microsporidial infection in
 415 immunocompetent individuals a longitudinal study. *PLoS Negl Trop Dis.* 2011; 5:
 416 e1162.
- 417 7. Ashikin A, Al-Mekhlafi HM, Moktar N et al. Molecular detection and species
 418 identification of *Enterocytozoon bieneusi* isolated from immunocompetent Orang
 419 Asli in Malaysia. *Parasitol Int.* 2017; 66:163–165.
- 420 8. Karimi K, Mirjalali H, Niyyati M et al. Molecular epidemiology of *Enterocytozoon*421 *bieneusi* and *Encephalitozoon* sp., among immunocompromised and
 422 immunocompetent subjects in Iran. *Microb Pathog*. 2020; 141: 103988.
- 423 9. Michlmayr D, de Sousa LA, Müller L et al. Incubation period, spore shedding
 424 duration, and symptoms of *Enterocytozoon bieneusi* genotype C infection in a

- 425 foodborne outbreak in Denmark, 2020. *Clin Infect Dis.* 2021; doi:
 426 10.1093/cid/ciab949.
- 427 10. Zhang Y, Koehler AV, Wang T et al. *Enterocytozoon bieneusi* of animals-With an
 428 'Australian twist'. *Adv Parasitol*. 2021; 111: 1–73.
- 429 11. Li W, Feng Y, Santin M. Host specificity of Enterocytozoon bieneusi and public
- 430 health implications. *Trends Parasitol.* 2019; 35: 436–451.
- 431 12. Wolfe N, Dunavan C, Diamond J. Origins of major human infectious diseases.
 432 *Nature*. 2007; 447: 279–283.
- 13. Trimmel NE, Walzer C. Infectious wildlife diseases in Austria-A literature review
 from 1980 until 2017. *Front Vet Sci.* 2020; 7: 3.
- 435 14. Acevedo P, Farfán MÁ, Márquez AL et al. Past, present and future of wild ungulates
 436 in relation to changes in land use. *Landscape Ecol.* 2011; 26: 19–31.
- 437 15. Martínez-Abraín A, Jiménez J, Jiménez I et al. Ecological consequences of human
 438 depopulation of rural areas on wildlife: A unifying perspective. Biol Conserv. 2020;
 439 252: 108860.
- 440 16. Carpio AJ, Apollonio M, Acevedo, P. Wild ungulate overabundance in Europe:
- 441 contexts, causes, monitoring and management recommendations. *Mam Rev.* 2021; 51:
 442 95–108.
- 443 17. Karim MR, Rume FI, Rahman ANMA et al. Evidence for zoonotic potential of
- 444 Enterocytozoon bieneusi in its first molecular characterization in captive mammals at
- Bangladesh National Zoo. *J Eukaryot Microbiol*. 2020; 67: 427–435.
- 18. Zhang Q, Zhong Z, Xia Z et al. Molecular epidemiology and genetic diversity of
 Enterocytozoon bieneusi in cervids from Milu Park in Beijing, China. Animals. 2022;
- 448 12: 1539.

- 449 19. Zhang K, Zheng S, Wang Y et al. Occurrence and molecular characterization of
- 450 *Cryptosporidium* spp., *Giardia duodenalis*, *Enterocytozoon bieneusi*, and *Blastocystis*
- 451 sp. in captive wild animals in zoos in Henan, China. *BMC Vet Res.* 2021; 17: 332.
- 452 20. Zhang Y, Koehler AV, Wang T et al. First detection and genetic characterisation of
- 453 *Enterocytozoon bieneusi* in wild deer in Melbourne's water catchments in Australia.
- 454 *Parasit Vectors*. 2018; 11: 2.
- 455 21. Li W, Deng L, Yu X et al. Multilocus genotypes and broad host-range of
 456 *Enterocytozoon bieneusi* in captive wildlife at zoological gardens in China. *Parasit*457 *Vectors*. 2016; 9: 395.
- 458 22. Amer S, Kim S, Han JI et al. Prevalence and genotypes of *Enterocytozoon bieneusi*459 in wildlife in Korea: a public health concern. *Parasit Vectors*. 2019; 12: 160.
- 23. Song Y, Li W, Liu H et al. First report of *Giardia duodenalis* and *Enterocytozoon bieneusi* in forest musk deer (*Moschus berezovskii*) in China. *Parasit Vectors*. 2018;
 11: 204.
- 463 24. Xie F, Zhang Z, Zhao A, Jing B, Qi M, Wang R. Molecular characterization of
- 464 Cryptosporidium and Enterocytozoon bieneusi in Père David's deer (Elaphurus
- *davidianus*) from Shishou, China. *Int J Parasitol Parasites Wildl.* 2019; 10: 184–187.
- 25. Zhang Z, Huang J, Karim MR et al. Zoonotic *Enterocytozoon bieneusi* genotypes in
 Pere David's deer (*Elaphurus davidianus*) in Henan, China. *Exp Parasitol*. 2015; 155:
- 468 46–48.
- 469 26. Huang J, Zhang Z, Yang Y et al. New genotypes of *Enterocytozoon bieneusi* isolated
 470 from sika deer and red deer in China. *Front Microbiol*. 2017; 8: 879.
- 27. Zhao W, Zhang W, Wang R et al. *Enterocytozoon bieneusi* in sika deer (*Cervus nippon*) and red deer (*Cervus elaphus*): deer specificity and zoonotic potential of ITS
- 473 genotypes. *Parasitol Res.* 2014; 113: 4243–4250.

- 474 28. Zhao W, Wang J, Yang Z et al. Dominance of the *Enterocytozoon bieneusi* genotype
- BEB6 in red deer (*Cervus elaphus*) and Siberian roe deer (*Capreolus pygargus*) in
- 476 China and a brief literature review. *Parasite*. 2017; 24: 54.
- 477 29. Liu W, Nie C, Zhang L et al. First detection and genotyping of Enterocytozoon
- 478 *bieneusi* in reindeers (*Rangifer tarandus*): a zoonotic potential of ITS genotypes.
- 479 *Parasit Vectors*. 2015; 8: 526.
- 30. Zhang XX, Cong W, Liu GH et al. Prevalence and genotypes of *Enterocytozoon bieneusi* in sika deer in Jilin province, Northeastern China. *Acta Parasitol.* 2016; 61:
 382–388.
- 31. Tao WF, Ni HB, Du HF et al. Molecular detection of *Cryptosporidium* and *Enterocytozoon bieneusi* in dairy calves and sika deer in four provinces in Northern
 China. *Parasitol Res.* 2020; 119: 105–114.
- 32. Santin M, Fayer R. *Enterocytozoon bieneusi*, *giardia*, and *Cryptosporidium* infecting
 white-tailed deer. *J Eukaryot Microbiol*. 2015; 62: 34–43.
- 488 33. Guo Y, Alderisio KA, Yang W et al. Host specificity and source of Enterocytozoon
- *bieneusi* genotypes in a drinking source watershed. *Appl Environ Microbiol*. 2014;
 80: 218–225.
- 34. Němejc K, Sak B, Květoňová D et al. Prevalence and diversity of *Encephalitozoon*spp. and *Enterocytozoon bieneusi* in wild boars (*Sus scrofa*) in Central Europe.
- 493 *Parasitol Res.* 2014; 113: 761–767.
- 494 35. Feng S, Jia T, Huang J et al. Identification of Enterocytozoon bieneusi and
- 495 Cryptosporidium spp. in farmed wild boars (Sus scrofa) in Beijing, China. Infect
 496 Genet Evol. 2020; 80: 104231.
- 497 36. Li W, Deng L, Wu K et al. Presence of zoonotic *Cryptosporidium scrofarum*, *Giardia*
- 498 *duodenalis* assemblage A and *Enterocytozoon bieneusi* genotypes in captive Eurasian

499 wild boars (*Sus scrofa*) in China: potential for zoonotic transmission. *Parasit Vectors*.

500 2017; 10: 10.

- 37. Lee H, Seo MG, Lee SH et al. Distribution and genotypic analysis of *Enterocytozoon bieneusi* from wild boars in Korea. *Med Mycol.* 2021; 59: 934–938.
- 503 38. Dashti A, Rivero-Juarez A, Santín M et al. *Enterocytozoon bieneusi* (Microsporidia):
- 504 Identification of novel genotypes and evidence of transmission between sympatric
- 505 wild boars (*Sus scrofa ferus*) and Iberian pigs (*Sus scrofa domesticus*) in Southern

506 Spain. *Transbound Emerg Dis.* 2020; 67:2869–2880.

- 39. González-Barrio D, Almería S, Caro MR et al. *Coxiella burnetii* shedding by farmed
 red deer (*Cervus elaphus*). *Transbound Emerg Dis*. 2015; 62: 572–574.
- 40. Thomas J, Balseiro A, Gortázar C et al. Diagnosis of tuberculosis in wildlife: a
 systematic review. *Vet Res.* 2021; 52: 31.
- 41. Caballero-Gómez J, Jiménez-Ruiz S, Lopez-Lopez P et al. Emergent subtype of
 hepatitis E virus genotype 3 in wild boar in Spain. *Transbound Emerg Dis.* 2019; 66:
 1803–1808.
- 42. Pérez-González J, Carranza J, Martínez R et al. Host genetic diversity and infectious
- diseases. Focus on wild boar, red deer and tuberculosis. *Animals (Basel)*. 2021; 11:
 1630.
- 517 43. PNVSFS, Plan Nacional de Vigilancia Sanitaria en Fauna Silvestre, MAPA,
- 518 Ministerio de Agricultura, Pesca y Alimentación. 2020. Retrieved from
- 519 <u>https://www.mapa.gob.es/es/ganaderia/temas/sanidad-animal-higiene-</u>
- 520 <u>ganadera/pvfs2020_tcm30-437517.pdf</u>. Accessed on May 17 2022.
- 44. Muñoz PM, Boadella M, Arnal M et al. Spatial distribution and risk factors of
 Brucellosis in Iberian wild ungulates. *BMC Infect Dis.* 2010; 10: 46.

- 45. Lorca-Oró C, López-Olvera JR, Ruiz-Fons F et al. Long-term dynamics of bluetongue
 virus in wild ruminants: relationship with outbreaks in livestock in Spain, 2006-2011. *PLoS One*. 2014; 9: e100027.
- 46. García-Bocanegra I, Paniagua J, Gutiérrez-Guzmán AV et al. Spatio-temporal trends

and risk factors affecting West Nile virus and related flavivirus exposure in Spanish

- 528 wild ruminants. *BMC Vet Res.* 2016; 12: 249.
- 47. Jiménez-Ruiz S, Vicente J, García-Bocanegra I et al. Distribution of Pestivirus
 exposure in wild ruminants in Spain. *Transbound Emerg Dis.* 2021; 68: 1577–1585.
- 48. González-Barrio D, Ortiz JA, Ruiz-Fons F. Estimating the efficacy of a commercial
- phase I inactivated vaccine in decreasing the prevalence of *Coxiella burnetii* infection
 and shedding in red deer (*Cervus elaphus*). *Front Vet Sci.* 2017; 4: 208.
- 18-month survey at a slaughterhouse in Massachusetts. *Appl Environ Microbiol*.
 2002; 68: 2595–2599.

49. Buckholt MA, Lee JH, Tzipori S. Prevalence of Enterocytozoon bieneusi in swine: an

- 537 50. Santín M, Fayer R. *Enterocytozoon bieneusi* genotype nomenclature based on the
 538 internal transcribed spacer sequence: a consensus. *J Eukaryot Microbiol*. 2009; 56:
- **539 34–38**.

527

- 540 51. Kimura M. A simple method for estimating evolutionary rates of base substitutions
- through comparative studies of nucleotide sequences. *J Mol Evol*. 1980; 16:111–120.
- 542 52. Kumar S, Stecher G, Li M et al. MEGA X: molecular evolutionary genetics analysis
 543 across computing platforms. *Mol Biol Evol.* 2018; 35: 1547.
- 544 53. R Core Team. (2012). R: A language and environment for statistical computing. R
- 545 Foundation for Statistical Computing, Vienna, Austria. ISBN 3-900051-07-0.
- 546 Retrieved from <u>http://www.R-project.org</u>. Accessed on May 17 2022.

- 54. Galván-Díaz AL, Magnet A, Fenoy S et al. Microsporidia detection and genotyping
 study of human pathogenic *E. bieneusi* in animals from Spain. *PLoS One*. 2014; 9:
 e92289.
- 55. Santín M, Calero-Bernal R, Carmena D et al. Molecular characterization of *Enterocytozoon bieneusi* in wild carnivores in Spain. *J Eukaryot Microbiol*. 2018;
 65:468–474.
- 553 56. Dashti A, Santín M, Cano L et al. Occurrence and genetic diversity of *Enterocytozoon bieneusi* (Microsporidia) in owned and sheltered dogs and cats in Northern Spain. *Parasitol Res.* 2019; 118:2979–2987.
- 556 57. Martínez-Padilla A, Caballero-Gómez J, Magnet Á et al. Zoonotic Microsporidia in
 557 wild lagomorphs in Southern Spain. *Animals (Basel)*. 2020; 10: 2218.
- 558 58. Haro M, Izquierdo F, Henriques-Gil N et al. First detection and genotyping of humanassociated microsporidia in pigeons from urban parks. *Appl Environ Microbiol*. 2005;
 71: 3153–3157.
- 561 59. Haro M, Henriques-Gil N, Fenoy S et al. Detection and genotyping of *Enterocytozoon* 562 *bieneusi* in pigeons. *J Eukaryot Microbiol*. 2006; 53 Suppl 1: S58–S60.
- 563 60. Vioque F, Dashti A, Santín M et al. Wild micromammal host spectrum of zoonotic
- eukaryotic parasites in Spain. Occurrence and genetic characterization. *Transbound Emerg Dis.* 2022 (in press).
- 566 61. Suarez VH, Busetti MR, Fort MC et al. Spiculopteragia spiculoptera, S. asymmetrica
- and Ostertagia leptospicularis from Cervus elaphus in La Pampa, Argentina. Vet
- 568 Parasitol. 1991; 40: 165–168.
- 569 62. Eve JH, Kellogg FE. Management implications of abomasal parasites in southeastern
 570 white-tailed deer. J Wildl Manage. 1977; 41: 169–177.

cervids, red deer, *Cervus elaphus* and fallow deer, *Dama dama*, from three localities
across central and western Spain: relationship to host density and park management.
J Parasitol. 2004; 90: 1378–1386.

63. Santín-Durán M, Alunda JM, Hoberg EP et al. Abomasal parasites in wild sympatric

575 64. Yang Y, Lin Y, Li Q et al. Widespread presence of human-pathogenic *Enterocytozoon*

- 576 *bieneusi* genotype D in farmed foxes (*Vulpes vulpes*) and raccoon dogs (*Nvctereutes*
- 577 procyonoides) in China: first identification and zoonotic concern. Parasitol Res.
 578 2015; 114:4341–4348.
- 579 65. Zhang XX, Cong W, Lou ZL et al. Prevalence, risk factors and multilocus genotyping
- of *Enterocytozoon bieneusi* in farmed foxes (*Vulpes lagopus*), Northern China. *Parasit Vectors*. 2016; 9: 72.
- 66. Xu C, Ma X, Zhang H et al. Prevalence, risk factors and molecular characterization
 of *Enterocytozoon bieneusi* in raccoon dogs (*Nyctereutes procyonoides*) in five
 provinces of Northern China. *Acta Trop.* 2016; 161: 68–72.
- 585 67. Leśniańska K, Perec-Matysiak A, Hildebrand J et al. Cryptosporidium spp. and
- 586 Enterocytozoon bieneusi in introduced raccoons (Procyon lotor)-first evidence from
- 587 Poland and Germany. *Parasitol Res.* 2016; 115: 4535–4541.
- 68. da Silva Fiuza VR, Lopes CW, de Oliveira FC et al. New findings of *Enterocytozoon bieneusi* in beef and dairy cattle in Brazil. *Vet Parasitol*. 2016; 216: 46–51.
- 590 69. ten Hove RJ, Van Lieshout L, Beadsworth MB et al. Characterization of genotypes
- of *Enterocytozoon bieneusi* in immunosuppressed and immunocompetent patient
 groups. *J Eukaryot Microbiol*. 2009; 56: 388–393.
- 593 70. Perec-Matysiak A, Leśniańska K, Buńkowska-Gawlik K et al. Zoonotic genotypes of
- 594 Enterocytozoon bieneusi in wild living invasive and native carnivores in Poland.
- 595 Pathogens. 2021; 10: 1478.

596

597 FIGURE LEGENDS

Fig. 1. Map of Spain showing the sampling areas and the geographical distribution of *Enterocytozoon bieneusi* DNA detected in wild and farmed ungulate species according to
established bioregions (BR1-5) as described in reference 44.

601

602 Fig. 2. Phylogenetic relationships among Enterocytozoon bieneusi complete ITS sequences (243 bp) generated in the present study (novel subtypes are represented with a 603 black filled circle and other subtypes with an unfilled circle) and representative reference 604 605 sequences for all E. bieneusi groups. PtEb XI genotype was used as outgroup to root the 606 tree. Analysis was conducted by a neighbor-joining method and genetic distances 607 calculated using the Kimura two-parameter model. Analysis involved 62 nucleotide sequences. Numbers at the nodes represent the bootstrap values with more than 50% 608 609 bootstrap support from 1000 pseudoreplicates.

610

611 **TABLE LEGENDS**

Table 1. Infection rates and molecular diversity of *Enterocytozoon bieneusi* reported in
wild, farmed, and captive ungulate (cervid and wild boar) species worldwide. Bolded
genotypes belong to zoonotic Group 1 and Group 2.

615

Table 2. Occurrence rates of *Enterocytozoon bieneusi* in wild free-ranging and farmed

617 ungulates (n = 1382) according to host species and established bioregions (BR1-5)

618 describes in reference 44.

620 Table 3. Frequency and molecular diversity of *Enterocytozoon bieneusi* genotypes621 identified in the wild and farmed ungulates investigated in the present study.

622

623 SUPPLEMENTARY TABLE LEGENDS

624 Supplementary Table 1. Number and relative frequencies of faecal samples from wild

and farmed ungulates (n = 1382) analysed in the present survey according to the

626 bioregion of origin.

627

628 Supplementary Table 2. Full dataset generated in the present study showing629 epidemiological, diagnostic, and genotyping results.

630

- 631 Supplementary Table 3. Occurrence rates and molecular diversity of *Enterocytozoon*
- *bieneusi* in wild and farmed ungulates according to the bioregion of origin.